Author’s response to reviews

Title: Transcatheter arterial chemoembolization plus sorafenib versus transcatheter arterial chemoembolization alone to treat advanced hepatocellular carcinoma: A meta-analysis

Authors:

Rong Cai (570039478@qq.com)
Rongfeng Song (929766813@qq.com)
Yan Yan (ronger2003011327@163.com)
Yifeng Liao (815354896@qq.com)
Cuiling Zhou (1536559823@qq.com)
Shuncong Wang (1624481793@qq.com)
Xiuling Zhou (987709137@qq.com)
Huaping Wang (344189537@qq.com)
Hongyu Zhang (zdwyzhy@163.com)
Haiqing Ma (mahaiqing@mail.sysu.edu.cn)
Huanhuan Sun (sunhuanh3@mail.sysu.edu.cn)
Pengfei Pang (785535164@qq.com)

Version: 3 Date: 27 Sep 2017

Author’s response to reviews:

Editor Comments:

1. We note the addition of a new author, Pengfei Pang, since the original submission of the manuscript. In line with COPE guidelines, BioMed Central requires written confirmation from all authors that they agree with any proposed changes in authorship of submitted manuscripts. Response 1: We have submitted a new authorship information form on September 23rd, 2017.
2. Please provide email addresses for all the co-authors on the title page.
Response 2: Thanks for your advice. In this document, we have already provided email addresses for all authors.
Rong Cai, E-mail: 570039478@qq.com
Rongfeng Song, E-mail: 785535164@qq.com
Pengfei Pang, E-mail: pflb@sina.com
Yan Yan, E-mail: yany8@mail2.sysu.edu.cn
Yifeng Liao, E-mail: leoifeng@163.com
Cuiling Zhou, E-mail: 1536559823@qq.com
Shuncong Wang, E-mail: 1624481793@qq.com
Xiuling Zhou, E-mail: 987709137@qq.com
Huaping Wang, E-mail: ronger2003011327@163.com
Hongyu Zhang, E-mail: zdwyzhy@163.com
Huanhuan Sun, E-mail: sunhuanh3@mail.sysu.edu.cn
Haiqing Ma, E-mail: mahaiqing@mail.sysu.edu.cn

3. Please name "Materials and methods" to "Methods" in the abstract and main body of your manuscript.
Response 3: We have substituted "Methods" for "Materials and methods" in the abstract and main body of our revised manuscript.

4. Please note that all manuscripts must contain all the following sections under the heading 'Declarations'. The Declarations should follow the Conclusions section, and be before the References.
Response 4: We thank you for the nice advice, and have made the “Declarations” section correct in the revised manuscript.

5. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.
Response 5: In the final version, we have submitted a clean version of our manuscript, including all relevant tables/figures/additional files.

6. Thank you for providing a copy of the language editing certificate. However, please remove it as an additional file as this is not needed at this stage.
Response 6: Thanks for your suggestion and we have removed the language editing certificate when submitting our manuscript this time.